Author's response to reviews

Title: Antibacterial activities of selected Cameroonian spices and their synergistic effects with antibiotics against multidrug-resistant phenotypes

Authors:

Aimé G. Fankam MSc (agfankam@yahoo.fr)
Victor Kuete Dr (kuetevictor@yahoo.fr)
Igor K. Voukeng MSC (tefogang@yahoo.fr)
Jules R. Kuiaté Dr (rkuiate@yahoo.com)
Jean-Marie Pages Dr (jean-marie.pages@univmed.fr)

Version: 3 Date: 21 September 2011

Author's response to reviews: see over
Cover Letter for the revised Manuscript MS: 6466984165818509

To the
Editor-in-Chief “BMC Complementary and Alternative medicine”

Via online system

Paper Submission

Dear editor

Thanks so much for sending to me the reviewers requirements for our accepted manuscript. Please find the answer for the point you requested for corrections

**Point 1.** Authors contributions have been provided in red color in the revised manuscript.
**Point 2.** All the clinical strains used are laboratory collections and were not directly provided from patient. They are well referred strains and no ethical approval for their use is needed.

Sincerely yours

Dr Victor Kuete